Author's response to reviews

Title: Adult attention deficit hyperactivity disorder is associated with asthma

Authors:

Ole Bernt Fasmer (ole.fasmer@kliniskmedisin.uib.no)
Anne Halmøy (anne.halmoy@biomed.uib.no)
Tomas Mikal Eagan (tomas.eagan@med.uib.no)
Ketil Joachim Oedegaard (ketil.oedegaard@helse-bergen.no)
Jan Haavik (jan.haavik@biomed.uib.no)

Version: 2 Date: 1 July 2011

Author's response to reviews:

Dear Prof Gerhard Andersson,

Thank you for your email (June 23, 2011) with comments on our paper “Adult attention deficit hyperactivity disorder is associated with asthma”. We have revised the manuscript according to the comments from the reviewer:

The reviewer states that: “It would have been nice if obesity data (BMI, waist circumference) were included, as well as smoking habits”.

We have already made a comment in the manuscript concerning the lack of data on smoking habits and obesity data. However we have made this more explicit, and mentioned lack of data on waist circumference in this connection (page 12, last paragraph, lines 23-24).

We hope that the manuscript now is acceptable for publication in BMC Psychiatry.

Yours,

Ole Bernt Fasmer